\begin{thebibliography}{10}

\bibitem{tongues-webpage}
{ACT II Tongues: Audio Voice Translation Guide System}.
\newblock http://www.avt-actii.lmowego.com/.

\bibitem{Burger}
S.~Burger, L.~Besacier, P.~Coletti, F.~Metze, and C.~Morel.
\newblock The nespole! voip dialogue database.
\newblock In {\em Proc. EUROSPEECH, European Conference on Speech Communication
  and Technology}, Aalborg, Denmark, September 2001. ISCA.

\bibitem{ASRU01}
R.~Cattoni, M.~Federico, and A.~Lavie.
\newblock Robust analysis of spoken input combining statistical and
  knowledge-based information sources.
\newblock In {\em Proceedings of ASRU-01}, December 2001.
\newblock to appear.

\bibitem{TiMBL:00}
Walter Daelemans, Jakub Zavrel, Ko~van~der Sloot, and Antal van~den Bosch.
\newblock {TiMBL: Tilburg Memory Based Learner, version 3.0 Reference Guide}.
\newblock Technical Report Technical Report 00-01, {ILK}, 2000.
\newblock Avaliable at {\small http://ilk.kub.nl/~ilk/papers/ilk0001.ps.gz}.

\bibitem{EEide}
E.~Eide.
\newblock Distinctive features for use in an automatic speech recognition
  system.
\newblock In {\em Proc. EUROSPEECH, European Conference on Speech Communication
  and Technology}, Aalborg, Denmark, September 2001. ISCA.

\bibitem{fuegen}
C.~F{\"u}gen and I.~Rogina.
\newblock Integrating dynamic speech modalities into context decision trees.
\newblock In {\em Proc. IEEE International Conference on Acoustics, Speech, and
  Signal Processing}, Istanbul, Turkey, June 2000. IEEE.

\bibitem{gates97}
Donna Gates, Alon Lavie, Lori Levin, Marsal Gavald\`{a}, Monika Woszczyna, and
  Puming Zhan.
\newblock End-to-end evaluation in janus: a speech-to-speech translation
  system.
\newblock In M.~Mast E.~Maier and S.~Luperfoy, editors, {\em Dialogue
  Processing in Spoken Language Systems}, Lecture Notes in Artificial
  Intelligence (1236), pages 195--206. Springer Verlag, 1997.

\bibitem{Gav00b}
M.~Gavald\`a.
\newblock Epiphenomenal {G}rammar {A}cquisition with {GSG}.
\newblock In {\em Proceedings of the {W}orkshop on {C}onversational {S}ystems
  of the 6th {C}onference on {A}pplied {N}atural {L}anguage {P}rocessing and
  the 1st {C}onference of the {N}orth {A}merican {C}hapter of the {A}ssociation
  for {C}omputational {L}inguistics ({ANLP}/{NAACL}-2000)}, Seattle, U.S.A, May
  2000.

\bibitem{Gol67}
{E. M.} Gold.
\newblock Language {I}dentification in the {L}imit.
\newblock {\em Information and Control}, 10(5), 1967.

\bibitem{workshop97}
Daniel Jurafsky, Rebecca Bates, Noah Coccaro, Rachel Martin, Marie Meteer,
  Klaus Ries, Elizabeth Shriberg, Andreas Stolke, Paul Taylor, and Carol~Van
  Ess-Dykema.
\newblock Switchboard discourse language modelling project final report.
\newblock Technical Report Research Note No. 30, Johns Hopkins University,
  Center for Speech and Language Processing, 1997 Large Vocabulary Continuous
  Speech Recognition Summer Research Workshop Technical Reports, 1998.

\bibitem{KKirchhoff}
K.~Kirchhoff.
\newblock Integrating articulatory features into acoustic models for speech
  recognition.
\newblock In {\em Proc. Phonus 5}, Saarbr{\"u}cken, Germany, 2001.
  Universit{\"a}t des Saarlandes.

\bibitem{Lavie2HLT}
A.~Lavie, C.~Langley, A.~Waibel, F.~Pianesi, G.~Lazzari, P.~Coletti, L.~Taddei,
  and F.~Balducci.
\newblock Architecture and design considerations in nespole!: a speech
  translation system for e-commerce applications.
\newblock In {\em Proc. Human Language Technology Conference}, pages 31--34,
  San Diego, CA, USA, March 2001. DARPA.

\bibitem{Lavie1HLT}
A.~Lavie, L.~Levin, T.~Schultz, C.~Langley, B.~Han, A.~Tribble, D.~Gates,
  D.~Wallace, and K.~Peterson.
\newblock Domain portability in speech-to-speech translation.
\newblock In {\em Proc. Human Language Technology Conference}, pages 82--86,
  San Diego, CA, USA, March 2001. DARPA.

\bibitem{lrec00}
Lori Levin, Boris Bartlog, Ariadna Font-Llitjos, Donna Gates, Alon Lavie,
  Dorcas Wallace, Taro Watanabe, and Monika Woszczyna.
\newblock Lessons learned from aa task-based evaluation of speech-to-speech
  machine translation.
\newblock In {\em Proceedings of LREC}, Athens, Greece, 2000.

\bibitem{icslp98}
Lori Levin, Donna Gates, Alon Lavie, and Alex Waibel.
\newblock {An Interlingua Based on Domain Actions for Machine Translation of
  Task-Oriented Dialogues}.
\newblock In {\em Proceedings of the International Conference on Spoken
  Language Processing (ICSLP'98)}, pages Vol. 4, 1155--1158, Sydney, Australia,
  1998.

\bibitem{mtj00}
Lori Levin, Alon Lavie, Monika Woszczyna, Donna Gates, Marsal Gavald\`{a},
  Detlef Koll, and Alex Waibel.
\newblock {The Janus-III Translation System}.
\newblock {\em Machine Translation}, 15(1-2):3--25, 2000.

\bibitem{LP00}
D.~Litman and S.~Pan.
\newblock Predicting and adapting to poor speech recognition in a spoken
  dialogue system.
\newblock In {\em Proc. of the Seventeenth National Conference on Artificial
  Intelligence (AAAI )}, Austin, TX, 2000.

\bibitem{soup}
{Marsal Gavalda}.
\newblock {SOUP: a Parser for Real-world Spontaneous Speech}.
\newblock In {\em Sixth International Workshop on Parsing Technologies}, pages
  101--110, Trento, Italy, February 2000.

\bibitem{Metze}
F.~Metze, J.~McDonough, and H.~Soltau.
\newblock Speech recognition over netmeeting connections.
\newblock In {\em Proc. EUROSPEECH, European Conference on Speech Communication
  and Technology}, pages 2389--2392, Aalborg, Denmark, September 2001. ISCA.

\bibitem{Nirenburg:92}
Sergei Nirenburg, Jaime Carbonell, Masaru Tomita, and Kenneth Goodman.
\newblock {\em {Machine Translation: A Knowledge-Based Approach}}.
\newblock Morgan Kaufmann, San Mateo, California, 1992.

\bibitem{Mostendorf}
M.~Ostendorf.
\newblock Moving beyond the beads-on-a-string model of speech.
\newblock In {\em Proc. IEEE Workshop on Automatic Speech Recognition and
  Understanding}, Keystone, Colorado, USA, 1999. IEEE.

\bibitem{Oviatt98}
S.~Oviatt.
\newblock The cham model of hyperarticulated adaptation during human-computer
  error resolution.
\newblock In {\em Proc. International Conference on Speech and Language
  Processing}, Sydney, Australia, November 1998. IEEE.

\bibitem{Qu97}
Yan Qu, Barbara~Di Eugenio, Alon Lavie, Lori Levin, and Carolyn Ros\'{e}.
\newblock {\em Minnimizing Cumulative Error in Discourse Context}.
\newblock Springer Verlag, Berlin, Heidelberg, New York, 1997.

\bibitem{Qu96}
Yan Qu, Carolyn Ros\'{e}, and Barbara~Di Eugenio.
\newblock Using discourse predictions for ambiguity resolution.
\newblock In {\em Proceedings of COLING}, 1996.

\bibitem{reithinger95}
Norbert Reithinger and Elizabeth Maier.
\newblock Using statistical dialogue act processing in verbmobil.
\newblock In {\em Proceedings of ACL}, 1995.

\bibitem{rose95}
C.~Ros\'{e}, B.~Di Eugenio, L.~Levin, and C.~Van Ess-Dykema.
\newblock Discourse processing of dialogues with multiple threads.
\newblock In {\em Proceedings of ACL}, 1995.

\bibitem{Rose98}
Carolyn Ros\'{e} and Lori Levin.
\newblock {An Interactive Domain Independent Approach to Robust Dialogue
  Interpretation}.
\newblock In {\em Proceedings of COLING/ACL}, 1998.

\bibitem{Soltau98}
H.~Soltau and A.~Waibel.
\newblock On the influence of hyperarticulated speech on recognition
  performance.
\newblock In {\em Proc. International Conference on Speech and Language
  Processing}, Sydney, Australia, November 1998. IEEE.

\bibitem{Suhm}
B.~Suhm, B.~Myers, and A.~Waibel.
\newblock Model-based and empirical evaluation of multimodal interactive error
  correction.
\newblock In {\em Proc. CHI Conference on Human Factors in Computing Systems},
  Pittsburgh, PA, USA, May 1999. ACM SIG.

\bibitem{kavita}
Kavita Thomas.
\newblock {Designing a Task-Based Evaluation Methodology for a Spoken Machine
  Translation System}.
\newblock In {\em Proceedings of ACL-99 (Student Session)}, College Park, MD,
  1999.

\bibitem{Takiguchi01}
T. Takiguchi, S.~Nakamura, and K. Shikano.
\newblock {HMM-Separation-based Speech Recognition for a Distant Moving
  Speaker}.
\newblock {\em {IEEE Transactions on Speech and Audio Processing}},
  9(2):127--140, 2001.

\bibitem{Walker01}
M.~Walker, J.~Aberdeen, J.~Boland, E.~Bratt, J.~Garofolo, L.~Hirschman, A.~Le,
  S.~Lee, S.~Narayanan, K.~Papineni, B.~Pellom, J.~Polifroni, A.~Potamianos,
  P.~Prabhu, A.~Rudnicky, G.~Sanders, S.~Seneff, D.~Stallard, and S.~Whittaker.
\newblock {DARPA} communicator dialog travel planning systems: The june 2000
  data collection.
\newblock In {\em Proceedings of Eurospeech}, 2001.

\bibitem{Walkeretal98}
M.~Walker, D.~Litman, C.~Kamm, and A.~Abella.
\newblock Evaluating spoken dialogue agents with {PARADISE}: Two case studies.
\newblock {\em Computer Speech and Language}, 1998.

\bibitem{walker97}
Marilyn Walker, D.~Litman, C.~Kamm, and A.~Abella.
\newblock {PARADISE: A Framework for Evaluating Spoken Dialogue Agents}.
\newblock In {\em Proceedings of the Annual Conference of the Association for
  Computational Linguistics (ACL'97)}, 1997.

\bibitem{WKL00}
Marilyn~A. Walker, Candace~A. Kamm, and Diane~J. Litman.
\newblock Towards developing general models of usability with {PARADISE}.
\newblock {\em Natural Language Engineering: Special Issue on Best Practice in
  Spoken Dialogue Systems}, 2000.

\bibitem{ward90}
Wayne Ward.
\newblock {The CMU Air Travel Information Service: Understanding Spontaneous
  Speech}.
\newblock In {\em Proceedings of the DARPA Speech and Language Workshop}, 1990.

\bibitem{Westphal}
M.~Westphal and A.~Waibel.
\newblock Model-combination-based acoustic mapping.
\newblock In {\em Proc. IEEE International Conference on Acoustics, Speech, and
  Signal Processing}, Salt Lake City, Utah, USA, May 2001. IEEE.

\bibitem{amta98}
Monika Woszczyna, Matthew Broadhead, Donna Gates, Marsal Gavald\`{a}, Alon
  Lavie, Lori Levin, and Alex Waibel.
\newblock {A Modular Approach to Spoken Language Translation for Large
  Domains}.
\newblock In {\em Proceedings of Conference of the Association for Machine
  Translation in the Americas (AMTA'98)}, Langhorn, PA, October 1998.

\end{thebibliography}

\section{Robust Speech Recognition}

In the context of a autonomous systems applicable in a wireless
environment the speech recognition engine is confronted with many
challenging conditions, such as varying environmental noise levels,
bandwidth limits, varying speaking styles, disfluencies, and speech
under stress, to name only a few.  To obtain acceptable performance,
current speech recognition systems need to be adapted to those
conditions. This adaptation requires large amounts of training data
that match the working conditions in question.

Until recently most ASR systems have been trained and used under
controlled conditions.  However, in the context of an immersive
environment such as an autonomous tourist assistant, these limitations
are unacceptable since systems should be easy to set-up and socially
acceptable, and should require little maintenance or attention. In
particular, 1) human users would rather not have to gear up with lapel
microphones, let alone headsets and headgear, or enrollment phases; 2)
humans will also want to talk freely without regard to gain settings or
start/stop buttons of the behind-the-scenes recognition technology; 3)
environmental noises should be filtered out to enhance recognition,
but also separately understood and tracked to model the environment.

We are planning to develop techniques and algorithms that allow for
robust speech recognition in the face of mismatched training and
testing conditions and build on the robust spoken language technology
of the {\sc Janus} project underway at our lab. We propose the
following approaches to achieve the goals of unrestricted open space
recognition for multiple languages:

\begin{enumerate}
\item Robust signal enhancement for remote and variable microphone positioning
\item Models of extreme speech, for example, under stress and with 
hyperarticulation
\item Using articulatory features in preparation for
language-independent speech recognition 
\end{enumerate}

\subsection{Robust Signal Enhancement for Remote and Variable Microphone Positioning}

The key issue in developing accurate recognition of noisy and
reverberant speech is to overcome the mismatch between training and
testing conditions. This problem is usually solved by the separate 
estimation of additive and convolutional noise in the linear and log 
spectrum domain, and then obtain feature compensation or model
adaptation through approximation. To address this, two classes of
algorithms have been developed. First is speech signal enhancement,
which can be considered as a form of unsupervised adaptation in the
signal feature domain. This type of algorithm is simple and efficient
but generally makes assumptions that limit their scope of
application. Second is acoustic model adaptation, which is usually
accomplished in two ways.  One way is direct adaptation of the HMM
parameters.  The other is indirect adaptation of a set of
transformation parameters followed by adaptation of the HMM parameters
using these transformations. Both methods are effective, but generally
slow to adapt or to compute.

Recently much promising new research has been conducted on model adaptation
\cite{Takiguchi01}. However these approaches are
computationlly costly and therefore only applicable for small, low
complexity tasks and are not scalable to LVCSR systems.

To overcome this limitation, we propose a new approach to combine the signal
enhancement and model adaptation algorithms together based on MAM
(model-combination-based acoustic mapping). Baseline MAM \cite{Westphal01} has
proved to be effective for additive noise. However, it only employs a simple
cepstral mean normalization for convolutional noise, which is not very
effective for remote microphone condition (YUE: OR HEAD MOVEMENTS?). In order
to compensate for reverberant speech, new adaptive algorithms need to be
developed. Our idea is to use a small size of secondary acoustic model for
robust signal enhancement. We can also utilize structure parameters in the
secondary model. Based on the secondary model, we can predict and map the
input signal distortion. Because it is a signal compensation methods, we avoid
modify the large size of acoustic models in the core speech recognition system
directly. It also has the advantage of on-line unsupervised model adaptation
methods, since we modify the secondary model to fit the current signal
condition. This adaptation process is fast since the size of secondary model
is small. The resulting system will be independent from microphone
positioning. The large size of acoustic models of core baseline clean speech
system will be trained on several available databases, namely BN (Broadcast
News), ESST (English Spontaneous Speech), and SWB (Telephone Speech)
corpus. Therefore it will be able model as many acoustic conditions as
possible. Although we can't cover all conditions, this will make the baseline
line system very robust and improve the overall speech recognition system
performance.

\subsection{Models of Extreme Speech}

To date, considerable progress has been achieved in speech recognition through
techniques such as vocal tract length normalization (VTLN), maximum likelihood
linear regression (MLLR) or speaker adaptive training. However, even in
dictation applications with 95\% word accuracy (WA), it is often necessary to
correct word errors. Since recognition systems always exhibit some errors, it
is important to examine how users react to recognition errors. In previous
studies \cite{Oviatt98},\cite{Soltau98}, it has been that shown that humans
follow similar recovery strategies as in interaction with humans when they use
recognition technology. In correction situations, speaking style becomes more
accentuated to disambiguate the original mistake. However, contrary to the
users intention, the word accuracy actually decreases significantly in case of
hyperarticulation. This is due to unmatched speaking style conditions and the
limitations of current phone based acoustic model approaches.

We propose to focus on building robust acoustics models to handle
hyperarticulated speech. We will extend our Janus system to build speech
recognizers that react more naturally on extreme speech input. The developed
techniques will be evaluated on different tasks to examine the robustness
against speaking style variations due to stress, emotion, hyperarticulation
and Lombard effect. The experiments will be carried out on different
databases, such as the public available SUSAS database (speech under simulated
and actual stress). An additional database, collected in our laboratories,
allows us to run comparison experiments between normal and hyperarticulated
speech.

We propose the following four steps:
\begin{enumerate}
\item Analysis of hyperarticulated effects \newline
In a first step, we plan to examine what kind of acoustic features (such as
speaking rate, pitch contour, formant frequencies) exhibit differences at
hyperarticulation. For example, the figures show the pitch contour for one
word spoken normally (left) and in a hyperarticulated mode (right).
\item Classifier to detect hyperarticulated speech \newline
In a second step, we plan to train statically classifiers to detect
hyperarticulation. The features for the classifier will be determined using
the results from work package 1.
\item Specialized acoustic models for normal / hyperarticulated speech \newline
After we have a classifier, we will train separate acoustic models for normal
and hyperarticulated speech (emotion-adaptive training). To that end, we can
use our JANUS speech recognition toolkit.
\item Early integration of hyperarticulated effects into context decision 
trees (Modalities) \newline 
To achieve a better parameter sharing, and to more
efficiently use available training data, we will study methods to incorporate
a ``hyperarticulation mode'' already in the context decision tree. The
advantage of such a method would be that exactly those acoustic models, which
exhibit differences at hyperarticulation, could be treated separately and the
phones, which do not need a special treatment, can share all speech data
together. The clustering procedure will determine if a model needs a
hyperarticulation mode automatically based on a cross-likelihood criterion.
\end{enumerate}

\subsection{Recognition Based on Articulatory Features and Processes}

Most current speech recognition systems represent speech as a linear sequence
of sound units, the so-called phones, with transitions between these units at
well-defined points in time.  During the training phase of a speech
recognizer, the acoustic characteristics of these phones are learned by
aligning the transcription with the speech waveform, assigning to each part of
the utterance exactly one phone ("beads-on-a-string model"
\cite{Mostendorf}. In reality, these clear-cut transitions between units do
not exist, as human speech production is a continuous process due to the
nature of our articulatory organs. To compensate for this, a typical
state-of-the-art LVCSR system represents a phone in context of its neighbours
and uses several thousands of these context-dependent units to model speech.

A wealth of compensation and adaptation schemes has been developed to improve
system behavior under adverse conditions such as background noise, cross-talk,
distant microphones, or difficult (narrow-band, distorting) transmission
channels (e.g. telephone or video-conferences) by adapting the models to the
new conditions. Because these adaptation techniques optimize the criterion
used during the training phase of the recognizer, they usually improve the
performance of the speech engine to a certain extent. However, these methods
do not really lead to robustness against changes in speaking style, e.g. from
planned to casual articulation. Initial research on the topic of spontaneous
speech has tried to compensate the effects of sloppy pronunciations by
expanding the dictionary with specific variations of important
entries. However, it is neither easy to generate new entries without
increasing the confusability nor is it clear how to balance variability
modeling through dictionary entries. Moreover, conventional pronunciation
modeling based on phones can not cover effects, where a realized hyper- or
under- articulated pronunciation of phone exhibits only certain articulatory
features, which do not correspond to a canonical phoneme.

We therefore propose to model phones not as a linear sequence of models, but
as an articulatory feature bundle, where each feature is in itself modeled by
a state sequence, but the transitions can occur at different times. For each
point in time the separate probability streams coming from the feature
detectors are combined by multiplying their probabilities. This
combination-of-streams approach has already been successfully applied to
noise-robust recognition, where different acoustic models have been trained
and it is assumed that they are differently affected by noise. Our approach
however applies this principle to feature detectors and additionally allows
the different streams to segment the data differently. This asynchronicity
distinguishes our system from other work e.g. presented in \cite{KKirchhoff}.

As a first step, we propose to limit the asynchronicity to one state and
determine the transitions for each stream separately simply by the
optimization of the training criterion (Maximum Likelihood in our case). The
goal of this phase is to train feature detectors, determine suitable features
streams and their respective stream weights, and to compare the computational
effort. The completion of this step will show the validity of the concept of
articulatory feature based ASR. We have already conducted initial experiments,
in which we combined six feature streams with a stream that simply consists of
our current 40k vocabulary English single-pass recognizer. The features
detectors were trained on a subset of the original training data only and
increased the size of the recognizer from approximately 133k Gaussians to
slightly less than 140k Gaussian models. The conventional HMM approach leads
to 14.1\% error rate and is significantly outperformed by adding the six
feature streams, which achieves 11.8\% error rate. These results were obtained
on a 20 minute in-house test-set comparable to the F0 condition of the
Broadcast News data. Other groups have reported gains on small systems using
feature-based approaches on noisy environments \cite{EEide}.

In a second step, we propose to use these stream weights for speaker and
channel adaptation and test the effectiveness of existing adaptation methods
with the aim of matching or improving the speech recognition performance of
the proposed articulatory feature stream method on a broad range of
tasks. Training, adapting and testing the systems on different in-house and
external tasks such as Broadcast News, Switchboard, and "Meeting" will
establish if the proposed method is indeed robust and works on a number of
tasks. The proposed system would allow for much finer modeling, by allowing
slow articulators to stay in place (alas skip states) while requiring fast
articulators (e.g. the velum) to make explicit transitions, therefore reducing
confusability with other words, which would otherwise increases word error
rate.

